Author's response to reviews

Title: An exploratory pilot study to assess self-perceived changes among social assistance recipients regarding employment prospects after receiving dental treatment

Authors:

Sonica Singhal (sonica.singhal@mail.utoronto.ca)
Muhammad Mamdani (MamdaniM@smh.ca)
Andrew Mitchell (andrew.mitchell4@sympatico.ca)
Howard Tenenbaum (Howard.Tenenbaum@dentistry.utoronto.ca)
Carlos Quiñonez (carlos.quinonez@utoronto.ca)

Version: 5 Date: 8 October 2015

Author's response to reviews:

Dear Editor,

The revised draft include all tables at the end of references of the main manuscript and the figure title/legend for figure 1 at the very end of the main manuscript. Tables have been removed from figure files.

Hope the manuscript is considered suitable from formatting perspective.

Regards,
Sonica